Author's response to reviews

Title: Can postponement of an adverse outcome be used to present risk reductions to a lay audience? A population survey

Authors:

Rasmus Dahl (r.dahl@dadlnet.dk)
Dorthe Gyrd-Hansen (dgh@dsi.dk)
Ivar Sonbo Kristiansen (ivar@c2i.net)
Jorgen Nexoe (jn@dadlnet.dk)
Jesper Bo Nielsen (jbnielsen@health.sdu.dk)

Version: 5 Date: 29 March 2007

Author's response to reviews:

The BioMed Editorial Production Team

Thank you for your letter of 27-03-2007.

We have made the following changes as requested:

1: Indicated the corresponding author with a symbol.
2: Listed all the authors names in reference 4.
4: Given the tables a section heading "Tables".
5: Ensured that Dorthe Gyrd-Hansen's name is identical in both the manuscript and submission system.

We hope the paper is now acceptable.

Yours sincerely

Rasmus Dahl
MD